Author’s response to reviews

**Title:** Modifiable risk factors for overweight and obesity in children and adolescents from Sao Paulo, Brazil

**Authors:**

- Scott Duncan (scott.duncan@aut.ac.nz)
- Elizabeth K Duncan (libduncan@gmail.com)
- Romulo A Fernandes (romulo_ef@yahoo.com.br)
- Camila Buonani (camilabuonani@yahoo.com.br)
- Karolynne DN Bastos (karolynne@fct.unesp.br)
- Aline FM Segatto (aline@fct.unesp.br)
- Jamil S Codogno (jamilecodogno@hotmail.com)
- Igor C Gomes (igorconterato@hotmail.com)
- Ismael F Freitas Jr (ismael@fct.unesp.br)

**Version:** 4 **Date:** 14 July 2011

**Author’s response to reviews:** see over
To the BMC Series Editorial Production Team

I have completed the major and minor changes to the proofs as requested and uploaded the edited manuscript. These changes include:

(1) Removal of all highlighting/tracking from the manuscript.
(2) Removal of line numbers.

There were no blank pages or typographical errors, so these changes were not made.

Please do not hesitate to contact me if there are any questions or concerns regarding this re-submission.

Kind regards

Dr Scott Duncan
Associate Director
Centre for Physical Activity and Nutrition
Auckland University of Technology
Private Bag 92006
Auckland, New Zealand
Phone: +649-921-9999 ext 7678
Fax: +649-921-9746
E-mail: scott.duncan@aut.ac.nz